Author's response to reviews

Title: Glucagonoma syndrome: a case report

Authors:

Pablo Granero Castro (pgranerocastro@aecirujanos.es)
Alberto Miyar de León (albertmiyar@yahoo.es)
Jose Granero Trancón (jose.granero@aecirujanos.es)
Paloma Álvarez Martínez (dra.palialvarez@gmail.com)
Jose A Álvarez Pérez (josealvar@telecable.es)
Jose C Fernández Fernández (blendio@terra.es)
Carmen M García Bernardo (cmgbernardo2@hotmail.com)
Luis Barneo Serra (lbarneo@uniovi.es)
Juan J González González (jjglezglez@gmail.com)

Version: 3 Date: 1 April 2011

Author's response to reviews: see over
The Biomed Central Editorial Team


Thank you for consideration of our manuscript for publication in your journal.
We have reviewed the above manuscript according to your review’s comments, noticing that both reviewers made the same considerations.

Reviewer #1 and #2 (Dr. Papavramidis and Dr. Pliakos)
COMMENTS TO AUTHORS:
1. Introduction; There are no references in the introduction section.
   • Some references were added in the introduction section.
2. Discussion; The first paragraph should be transferred to the introduction.
   • Done. Paragraph transferred to page 1.
3. Discussion; The second paragraph should be deleted, it offers nothing to the manuscript.
   • Done.
4. Discussion; The third paragraph should be transferred to the introduction.
   • Done. Paragraph transferred to page 1.
5. Discussion; The 5th paragraph should be the first paragraph of the discussion since this is a case report orientated to the glucagonoma syndrome.
   • Done.
• A sentence concerning the differential diagnosis of glucagonoma has been added in paragraph 5 of the Discussion (page 6) along with its corresponding references.

7. **Discussion**: There should be a paragraph concerning the rare pancreatic tumors [Padhi JOP 2010] [Espinal-Witter Virchows Archives 2010] [Cheng Cancer Treat Res 2010] [Papavramidis Annals Gastroenterol 2010] [Apostolidis J Med Case Reports 2008].

• A sentence concerning the rare pancreatic tumours has been included in the paragraph 5 of the Discussion (page 7) along with its corresponding references.

**FORMATTING CHANGES:**

8. Please include the ethnicity of the patient in the case presentation section of the manuscript.

• Done (first paragraph, page 2).

9. If applicable, please include an acknowledgement section at the end of the manuscript before the reference list.

• No acknowledgement section is required in this manuscript.